Author's response to reviews

Title: A IGF-I promoter polymorphism modifies the relationship between birth weight and risk factors for cardiovascular disease and diabetes at age 36

Authors:

Saskia J te Velde (saksiavelde@hotmail.com)
Elisabeth FC van Rossum (e.vanrossum@erasmusmc.nl)
Paul G Voorhoeve (pg.voorhoeve@vumc.nl)
Jos WR Twisk (jwr.twisk@vumc.nl)
Henriette A Delemarre-van de Waal (H.delemarre@vumc.nl)
Coen DA Stehouwer (cda.stehouwer@vumc.nl)
Willem van Mechelen (w.vanmechelen@vumc.nl)
Steven WJ Lamberts (s.w.j.lamberts@erasmusmc.nl)
Han CG Kemper (hcg.kemper@vumc.nl)

Version: 5 Date: 31 May 2005

Author's response to reviews:

Dear Sally Dennisson, assistant editor,

Thank you for your email mentioning the acceptance of our manuscript 'A IGF-I promoter polymorphism modifies the relationship between birth weight and risk factors for cardiovascular disease and diabetes at age 36'.

According to the requested formatting changes, we changed the following:
- in the authors list Henriette A Delemarre - van de Waal
- running title is removed
- word count is removed
- keywords are removed
- 'design and methods' is changed in 'methods'
- the suggested table position indicators are removed
- a conclusions section is added
- table legends are removed, they are still with the tables
- additional files are removed

We hope the publishing process can continue now.

Best wishes

Saskia te Velde

Dept. of Public Health
Erasmus MC
University Medical Center Rotterdam
P.O. Box 1738
3000 DR Rotterdam
The Netherlands
Phone: +31-10-408 77 18
Fax : +31-10-408 94 49
Email: s.tevelde@erasmusmc.nl